Title: Association between registered nurse staffing levels and in-hospital mortality in craniotomy patients using Korean National Health Insurance data

Authors:
Yunmi Kim (kyunm@eulji.ac.kr)
SeYoung Kim (sarakimk@changwon.ac.kr)
Kyounga Lee (tj720221@snu.ac.kr)

Version: 4 Date: 26 Apr 2020

Author’s response to reviews:

Thank you very much for your helpful comments.

1. Section: Please assign your manuscript to the relevant section.
   -&gt; We assigned this manuscript to the “Service Organization, Utilization, and Collaborative Care” section.

2. Authors Contributions: please double check the initials used here, currently, there are confusing.
   -&gt; We checked and corrected the initials. YK is Yunmi Kim, SK is SeYoung Kim, and KL is Kyounga Lee.

3. Availability of Data and Materials: Please specify who to contact, for example, the corresponding author on reasonable request with permission from XXX.
   -&gt; We revised the “Availability of Data and Materials” using the statement form which is suggested in your editorial policies (http://www.biomedcentral.com/submissions/editorial_policies#availability+of+data+and+materials)
   “The data that support the findings of this study are available from the National Health Insurance Service (NHIS) but restrictions apply to the availability of these data, which were used under license for the current study, and so are not publicly available. Data are however available from the authors upon reasonable request and with permission of the NHIS.”

4. Copy edit: Your manuscript needs to be copyedited. We recommend that you ask a native English speaking colleague to help you copyedit the paper. If this is not possible, you may need to use a professional language editing service. For authors who wish to have the language in their manuscript edited by a native-English speaker with scientific expertise, BioMed Central has a new in house editing service. The new editing tool can provide both scientific and language editing: http://authorservices.springernature.com/
   -&gt; Throughout the process of writing and revising this article, copyediting was carried out using a professional language editing service. In this final manuscript, additional changes to punctuation and grammar have been made, reflecting comments provided by previous reviewers regarding fine points of phrasing, content-specific word and style. We hope that this manuscript meets your expectations.
5. Formatting:
1) Please remove the corresponding author section from the declarations.
   -&gt; We have removed this section.
2) Please remove the response to reviewers comments.
   -&gt; We have removed the responses.
3) Please ensure that all figures/tables and supplementary files are cited within the text. Any items which are not cited may be deleted by our production department upon publication.
   -&gt; We have ensured that all figures/tables are cited within the text.

6. Clean: At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files. Please ensure that all figures, tables and additional-supplementary files are cited within the text.
   -&gt; We have uploaded a final clean manuscript file and a separate figure file as you suggested.